Author's response to reviews

Title: The Six Minute Walk Test Accurately Estimates Mean Peak Oxygen Uptake

Authors:

Robert M Ross (rross@bcm.tmc.edu)
Jayasimha N Murthy (Jayasimha.murthy@uth.tmc.edu)
Istvan D Wollak (iwollak@citomail.hu)
Andrew S Jackson (udde@mac.com)

Version: 6 Date: 20 May 2010

Author's response to reviews:

Major revisions
- Adjusted equations to be less than 85 characters.
- Added that all authors read and approved the final manuscript.
- fixed reference problem.

Minor revisions
- formatted authors contributions as a paragraph.
- Left other changes to for you.

Thank you